\begin{table}%t2
\caption{Line luminosities.}\label{tab2}
%\centerline
{
\begin{tabular}{lllcc} \hline \hline
No.& Lines & Reference &log Flux (\flux) & log Luminosity (\lum)\\
(1) &~ (2)   &~(3)  &(4) & (5)\\ \hline
1 &\civ, \lya                 &15          &$-$11.74&  45.22\\
2 &\ha, \hb, \hg              &1           &$-$11.42&  45.09\\
3 &\hg, \civ, \mgii, \lya     &16          &$-$11.49&  45.34\\
4 &\hb,\hg, \mgii             &5           &$-$11.93&  44.95\\
5 &\civ, \lya                 &7,11        &$-$11.41&  44.96\\
6 &\ha                        &6           &$-$12.60&  44.21\\
7 &\ha, \hb, \civ, \mgii, \lya&7,9,10,11,12&$-$11.21&  46.03\\
8 &\hb, \civ, \lya            &9,11        &$-$11.64&  45.53\\
9 &\ha, \hb, \hg, \civ, \lya  &8,11        &$-$11.80&  44.43\\
10&\ha, \hb, \hg, \civ, \lya  &8,10,11     &$-$11.64&  44.63\\
11&\hg, \mgii                 &16          &$-$11.95&  44.98\\
12&\ha, \hb, \civ, \mgii, \lya&8,10,11     &$-$11.67&  44.77\\
13&\ha, \hb, \hg              &13          &$-$11.79&  44.47\\
15&\ha, \hb, \hg, \civ, \lya  &8,11,7      &$-$11.87&  44.57\\
18&\hb, \hg, \civ, \mgii, \lya&3,14        &$-$10.28&  45.79\\
20&\ha, \hb, \hg, \civ, \mgii &8,15        &$-$11.96&  44.75\\
21&\hb, \civ, \lya            &7,9,11      &$-$11.51&  45.09\\
22&\civ, \lya                 &11          &$-$11.50&  44.70\\
23&\ha, \hb, \civ, \mgii, \lya&10,11       &$-$11.66&  45.16\\
24&\ha, \hb, \civ, \mgii, \lya&7,9,10,11   &$-$11.59&  45.25\\
25&\hb, \civ, \mgii, \lya     &7,9,15      &$-$11.11&  45.42\\
26&\hb, \hg                   &2           &$-$12.34&  44.64\\
27&\hb, \hg, \civ, \lya       &8,11        &$-$11.84&  45.44\\
28&\ha, \hb, \civ, \mgii, \lya&7,9,10,11   &$-$11.63&  45.21\\
29&\hb, \hg, \civ, \mgii, \lya&2,7,11,12   &$-$12.04&  45.08\\
30&\hb, \civ, \lya            &7,9,11      &$-$11.40&  44.87\\
31&\hb, \civ, \lya            &7,9,11      &$-$11.69&  44.64\\
32&\ha, \hb, \hg, \civ, \lya  &7,8,9,10,11 &$-$11.31&  45.31\\
33&\hb                        &4           &$-$12.22&  44.21\\
35&\hb                        &4           &$-$11.25&  44.89\\ \hline
\end{tabular}}
\par
\medskip
{\sc Notes --}
Colomn (1) number; Col.~(2) emission lines; Col.~(3) references; Col.~(4) the
derived total line fluxes; and Col.~(5) the derived total line luminosity.\\
{\sc References --}
 (1) Baldwin (\cite{Baldwin75});
 (2) Bergeron \& Kunth (\cite{Bergeron84});
 (3) Corbin (\cite{Corbin92});
 (4) Corbin (\cite{Corbin97});
 (5) Danziger \& Goss (\cite{Danziger83});
 (6)~Eracleous \& Halpern (\cite{Eracleous94});
 (7) Gondhalekar (\cite{Gondhalekar90});
 (8) Jackson \& Browne (\cite{Jackson91});
 (9) Marziani et al. (\cite{Marziani96});
(10) Neugebauer et al. (\cite{Neugebauer79});
(11) Osmer et~al. (\cite{Osmer94});
(12) Oke et~al. (\cite{Oke84});
(13) Scarpa \& Falomo (\cite{Scarpa97});
(14) Stockton \& MacKenty (\cite{Stockton87});
(15) Wills et al. (\cite{Wills95});
(16) Wilkes et al. (\cite{Wilkes99}).
\end{table}